Author’s response to reviews

Title: Can dental CT predict the quality of interradicular bone?

Authors:

Tamar Brosh (tbrosh@post.tau.ac.il)
Yekaterina Bereznjak- Elias (katybreznik@yahoo.com)
Rafael Pilo (rafipilo@post.tau.ac.il)
Nir Shpack (nir@shpack.co.il)
Silvia Geron (sigeron@zahav.net.il)

Version: 2 Date: 27 December 2013

Author’s response to reviews:

In this revised manuscript, we added the title page, conclusions section, keywords (after the abstract), list of abbreviations, competing interests' declaration, authors' contributions and acknowledgements.